Author’s response to reviews

Title: Patient prioritization tools and their effectiveness in non-emergency healthcare services: A systematic review protocol

Authors:

Julien Dery (julien.dery.4@ulaval.ca)
Angel Ruiz (angel.Ruiz@fsa.ulaval.ca)
Francois Routhier (francois.routhier@rea.ulaval.ca)
Marie-Pierre Gagnon (marie-pierre.gagnon@fsi.ulaval.ca)
Andre Cote (andre.Cote@fsa.ulaval.ca)
Daoud Ait-Kadi (daoud.Aitkadi@gmc.ulaval.ca)
Valerie Belanger (valerie.3.belanger@hec.ca)
Simon Deslauriers (simon.deslauriers.1@ulaval.ca)
Marie-Eve Lamontagne (marie-eve.lamontagne@cirris.ulaval.ca)

Version: 1 Date: 11 Jan 2019

Author’s response to reviews:

We first want to really thank the very educative and positive reviews received. As the first author of this protocol is a graduate student, this publication exercise was an important asset for his studies and, with the authorization of the editor, I'd like to use the reviewers' comments for education purpose in graduate class as an example of positive and helpful review. Thank you again for the support and thoughtful comments that were appreciated by the whole research team.